Author's response to reviews

Title: International students' experience of a western medical school: A mixed methods study exploring the early years in the context of cultural and social adjustment compared to students from the host country

Authors:

Alice McGarvey (amcgarvey@rcsi.ie)
Ruairi Brugha (rbrugha@rcsi.ie)
Ronan Conroy (rconroy@rcsi.ie)
Eric Clarke (eclarke@rcsi.ie)
Elaine Byrne (ebyrne2@rcsi.ie)

Version: 4 Date: 9 April 2015

Author's response to reviews:

Dear Rose

Thank you for your email, received 8th April 2015. As requested the information regarding ethical approval has been altered and the revised manuscript uploaded.

Sincerely

Alice McGarvey